Author’s response to reviews

Title: Combined antibacterial effect of essential oils from three most commonly used Ethiopian traditional medicinal plants on multidrug resistant bacteria

Authors:

Eshetu Dabale (gadisaeshetu@gmail.com)
Gebru Weldeargay (gebrumulu@gmail.com)
Kassu Tulu (kassudesta@gmail.com)
Getahun Dibaba (getahunthegaye@gmail.com)
Sintayehu Ayena (sintayehu1981@gmail.com)
Kefiyelw Duffera (naftan2006@gmail.com)
Abera Dereressa (aberatekale91@gmail.com)

Version: 4 Date: 29 Dec 2018

Author’s response to reviews:

Dear Dr.Emily Wollmuth

Thank you for your consideration our research in such highly qualified and reputable journal. We were incorporated all things you recommended us to do so.

Once again I want to say thanks a lot!